Author's response to reviews

Title: Allele-specific disparity in breast cancer

Authors:

Fatemeh Kaveh (fateka@rr-research.no)
Hege Edvardsen (Hege.Edvardsen@rr-research.no)
Anne-Lise Børresen-Dale (a.l.borresen-dale@medisin.uio.no)
Vessela N Kristensen (v.n.kristensen@medisin.uio.no)
Hiroko K Solvang (hsolvang@rr-research.no)

Version: 3 Date: 29 November 2011

Author's response to reviews: see over
November 28, 2011

Journal Editorial Office  
BioMed Central

Dear Elaine Cruz:

I’m sending the revised version of our manuscript (MS: 3658088005958297) entitled “Allele-specific disparity in breast cancer” for your consideration for publication in BMC Medical Genomics.

We thank the reviewers for their thorough evaluation and excellent suggestions. We have addressed all of the comments raised by both reviewers, Professor Wei Sun and Dr. Kun Huang, and incorporated their input in the revised manuscript.

According to Dr. Kun Huang’s comments, we separated Results and Discussion. We also modified the titles of figures 2, 6 and the missing parts in the flowchart of supplementary figure 1. Reference literatures were also modified.

Based on Professor Wei Sun’s comment, we simplified figure 1.

Regarding the reviewers comments, we have addressed and highlighted the revised parts in both the manuscript and our responses.

We hope that the revised version will meet the journal’s requirements and that you will approve it for publication. Thank you for your patience and guidance.

Sincerely yours,

Fatemeh Kaveh, PhD student

Department of Genetics  
Institute for Cancer Research  
Oslo University Hospital Radiumhospitalet  
Montebello, 0310 Oslo, Norway  
Phone: +47 22 78 13 68  
E-Mail: fateka@rr-research.no